Author's response to reviews

Title: Gastroesophageal reflux symptoms in infants in a rural population: longitudinal data over the first six months

Authors:

   Robert S Van Howe (vanhowe@miuplink.net)
   Michelle R Storms (mrstorms@mgh.org)

Version: 5 Date: 26 January 2010

Author's response to reviews:

Dear Editors,

The recommended formatting changes have been made. Please contact me if you have any questions.

Sincerely,

Robert S. Van Howe, MD, MS, FAAP